

# This AFME Standard Form Agreement Among Underwriters speaks as of 1 January 2021.

For use in connection with an equity offering involving a prospectus (although it may be adapted very easily for an undocumented issuer capital raising as needed) – currently drafted for use with a pathfinder/preliminary prospectus in the context of an IPO; but with minor edits it can be adapted for use on rights issues or add-on offerings, whether with a final prospectus only or with both a pathfinder/preliminary plus final prospectus. This standard form has been drafted for issuer's shares being admitted to trading on the main market of the London Stock Exchange but it is intended to be used (with minor adaptations), for equity offerings being listed on any other stock exchange throughout EMEA. Drafting options in relevant clauses provide for both a UK onshored version of and the original EU version of certain EU directives/regulations relevant to Underwriters' activities, such as MAR, MiFID II and BRRD. Facts and circumstances such as the jurisdiction of the Issuer, each Underwriter and the relevant listing venue will determine which option is to be used.

**THIS AGREEMENT** is dated [ ] 20[]

#### **BETWEEN:**

- (1) **[GLOBAL CO-ORDINATOR]** of **[** ] (*Global Co-ordinator One*);
- (2) **[GLOBAL CO-ORDINATOR]** of **[** ] (*Global Co-ordinator Two* and, together with Global Co-ordinator One, the *Joint Global Co-ordinators*);
- (3) **[UNDERWRITER ONE]** of **[** ] (*Underwriter One* and, together with the Joint Global Coordinators, the *Joint Bookrunners*);
- (4) [UNDERWRITER TWO] of [ ] (Underwriter Two); and
- (5) **[UNDERWRITER THREE]** of [] (*Underwriter Three* and, together with Underwriter Two, the *Co-Lead Managers*, and the Co-Lead Managers together with the Joint Bookrunners, the *Underwriters*).

## WHEREAS:

- (A) In connection with the proposed offering of Shares (the *Offer*) in [] plc (the *Company*), the Company has issued a pathfinder prospectus dated [] 20[] (the *Pathfinder Prospectus*) and is proposing to issue a final prospectus (the *Prospectus*, and together with the Pathfinder Prospectus and any supplementary prospectus¹ published by the Company pursuant to Article 23 of the [EU] [UK] Prospectus Regulation, the *Offering Circulars*) dated the same date as this Agreement stating, amongst other things, the price at which Shares are being offered pursuant to the Offer.
- (B) The Company, [the Directors,] [the Selling Shareholders,] [the Sponsor] <sup>2</sup> and the Underwriters have entered into an underwriting [and sponsor's] agreement dated [] 20[] (the *Underwriting Agreement*) pursuant to which:

-

<sup>[</sup>Note for UK deals - Although the UK Prospectus Regulation, in Article 23, uses the term "supplements", Rule 3.4 of the Prospectus Regulation Rules refers to "supplementary prospectus". Regardless of the use of the term supplements in the UK Prospectus Regulation, UK market practice is still to use the term "supplementary prospectus".

UK-centric elements of the agreement are highlighted in yellow.

- (a) each of the Underwriters has agreed, severally and not jointly or jointly and severally, that the Joint Bookrunners<sup>3</sup> will seek to procure:
  - (i) as agents for the Company, subscribers for the number of New Shares shown opposite its name in column [] of the table in Part [] of schedule [] to the Underwriting Agreement, [and
  - (ii) as agents for the Selling Shareholders, purchasers for the number of Sale Shares shown opposite its name in column [] of the table in Part [] of Schedule [] to the Underwriting Agreement; and
- (b) each of the Underwriters has agreed, severally and not jointly or jointly and severally, that:
  - (i) if and to the extent that subscribers fail to be procured for or to subscribe for all of the New Shares (such number of New Shares which are not so subscribed being the *Unplaced New Shares*), it will subscribe for its Agreed Proportion of the Unplaced New Shares; [and
  - (ii) if and to the extent that purchasers fail to be procured for or to purchase all of the Sale Shares (such number of Sale Shares which are not so purchased being the *Unplaced Sale Shares* and, together with the Unplaced New Shares, the *Unplaced Shares*), it will purchase its Agreed Proportion of the Unplaced Sale Shares,

in each case,] at the Offer Price on the Closing Date and on the terms and subject to the conditions set out, or referred to, in the Underwriting Agreement.

- (C) In addition, pursuant to the Underwriting Agreement, [], in its capacity as Stabilising Manager, has [entered into an agreement for the sale and purchase of the Over-allotment Shares with the Over-allotment Shareholders pursuant to which the Stabilising Manager may purchase, or procure purchasers for, up to a maximum of [] Shares], on the terms and subject to the conditions set out, or referred to, in the Underwriting Agreement.
- (D) Each of the Company[, the Selling Shareholders] [and the Directors] has appointed [] and [] as joint global co-ordinators for the purposes of co-ordinating the Offer.
- (E) [Each of the Company[, the Selling Shareholders] [and the Directors] has appointed [], [] and [] as joint bookrunners for the purposes of the Offer.]<sup>4</sup>
- (F) [The Shares will be made available in the United States to qualified institutional buyers in accordance with Rule 144A under the Securities Act, or pursuant to another available exemption from, or in a transaction not subject to, the registration requirements of the Securities Act, and to institutional and certain other investors outside the United States in reliance on Regulation S under the Securities Act.]

## IT IS AGREED as follows:

#### 1. INTERPRETATION

1.1 The Recitals and the Schedule to this Agreement have effect as parts of this Agreement.

Include reference to the Underwriter(s) with the primary decision-making function. Whether it is the bookrunners or the global co-ordinators that drive the syndicate is likely to differ according to the transaction in question. In particular, where the bookrunners and the global co-ordinators are not the same entities, the division of responsibilities and the decision-making split between the two roles will need to be agreed and reflected in the Agreement. This standard form Agreement has been drafted on the assumption that the Joint Bookrunners have the primary decision-making function and references to the Joint Bookrunners are, therefore, included throughout. If this is not the case, the Agreement will need to be amended accordingly.

See footnote 3 above.

- 1.2 This Agreement (and the Recitals and the Schedule) is to be construed and interpreted in accordance with Schedule 1.
- 1.3 Each word and expression used in this Agreement (including in the Recitals and the Schedule) has, except insofar as the context requires otherwise or unless defined elsewhere in this Agreement, the meaning (if any) given to it in Schedule 1 or in the Underwriting Agreement. Insofar as there is any conflict between a meaning given to a word or expression by both this Agreement and the Underwriting Agreement, the meaning given by this Agreement shall prevail.

#### 2. AUTHORITIES

- 2.1 Each of [Underwriter Two] and [Underwriter Three] (together, the *Non-JBR Underwriters* and each a *Non-JBR Underwriter*)<sup>5</sup> hereby irrevocably appoints each of the Joint Bookrunners as its agent to take on its behalf separately or together such actions as the Joint Bookrunners may consider necessary (acting in good faith) to give effect to the transactions contemplated by the Underwriting Agreement, this Agreement and any ancillary agreements or arrangements, including (without limitation):
- (a) determining the allocation of the Offer Shares among investors pursuant to the Offer;
- (b) agreeing any addition or amendment to, or waiver or modification of, any provision of the Underwriting Agreement (except such Non-JBR Underwriter's Underwriting Commitment (other than pursuant to Clause 4)), if and to the extent not materially prejudicial to the Non-JBR Underwriters in the good faith opinion of the Joint Bookrunners<sup>6</sup>;
- (c) appointing any appropriately regulated bank, broker or dealer to procure or to assist in procuring subscribers [and/or purchasers] for such number of the Offer Shares as the Joint Bookrunners may determine;
- (d) waiving (in whole or part) compliance with, or satisfaction of, any of the conditions or obligations referred to in the Underwriting Agreement (including, for the avoidance of doubt, any of the Conditions set out in clause [] of the Underwriting Agreement), granting any extension of any time limit or period fixed under the Underwriting Agreement (provided that the Joint Bookrunners will not agree to any amendment to the definition of Long Stop Date without the agreement of all of the Non-JBR Underwriters) and/or making any determination falling to be made by the Underwriters (or any one or more of them) under the Underwriting Agreement or in connection with it;
- (e) following any default by an Underwriter in its obligation to pay or procure payment for the number of Shares required by Clause 3.4 and Clause 6, determining the proportions in Clause 4 on the basis determined by the Joint Bookrunners;
- (f) in the case of the Stabilising Manager [(with the consent of the other Joint Bookrunners)]<sup>7</sup>, giving any notice for which the Underwriting Agreement provides in relation to the Over-allotment Shares;
- (g) entering into any arrangements that the Joint Bookrunners consider appropriate to effect payment for, and delivery of, the Offer Shares in connection with the Offer; [and]

See footnote 3 above. If, in any particular case, the Underwriter(s) with the primary decision making function are the global coordinators, change the initial part of this paragraph to read, "Each of [Underwriter One], [Underwriter Two] and [Underwriter Three] (together, the Non-GC Underwriters and each a Non-GC Underwriter)..." and change all references to "Non-JBR Underwriters"/"Non-JBR Underwriters"/"Non-JBR Underwriter" to "Non-GC Underwriters"/"Non-GC Underwriter" respectively.

To be amended as necessary if different entities are acting as bookrunners and global co-ordinators.

To be consistent with the approach agreed between the Stabilising Manager and the Joint Bookrunners. See also Clause 8.2.

- (h) executing and delivering, on the basis determined by the Joint Bookrunners, the Underwriting Agreement on behalf of themselves and each of the Non-JBR Underwriters<sup>8</sup>; [and]
- (i) [include a reference to any other authorities, if any, that need to be specifically mentioned.]
- 2.2 Each Underwriter agrees to ratify and confirm all and any action taken lawfully and in good faith on its behalf by the Joint Bookrunners or any of them in the exercise of the powers granted by Clause 2.1 and/or the other clauses of this Agreement and shall be bound by the terms of the Underwriting Agreement as executed or amended in accordance herewith.
- 2.3 None of the Joint Bookrunners shall have any responsibility or liability to any of the other Underwriters in relation to any exercise or failure to exercise a power granted by Clause 2.1 or in relation to any other exercise of its discretion under the Underwriting Agreement or this Agreement or in relation to its role as a Joint Bookrunner so long as it has acted in good faith. No obligations not expressly assumed by the Joint Bookrunners in this Agreement or the Underwriting Agreement shall be implied herefrom or therefrom.
- 2.4 Each Underwriter confirms that it has examined the Offer Documents published on or prior to the date of this Agreement, each as amended to date, and is familiar with the rights and restrictions attaching to the Shares and the terms of the Offer, including the over-allotment arrangements as described in the Prospectus. Each Underwriter confirms that the information relating to it in those documents is correct. The Joint Bookrunners are authorised to consent, at their discretion on behalf of themselves and each of the other Underwriters, to any further amendments or supplements to the Offer Documents or any of them as may be agreed with the Company, provided that, in the opinion of the Joint Bookrunners, acting in good faith, any such further amendments or supplements are not materially prejudicial to the Non-JBR Underwriters.

# 3. BOOKBUILDING AND OFFERING

- 3.1 Each Underwriter agrees that:
- (a) the Offer Price has been/will be agreed between the Company and the Joint Bookrunners (who, for the avoidance of doubt, are acting for themselves and on behalf of each of the other Underwriters), having regard to the outcome of the bookbuilding process;
- (b) allocations among investors pursuant to the Offer will be established by the Joint Bookrunners in their absolute discretion, subject to any contractual requirement pursuant to the Underwriting Agreement to agree those matters with the Company [and the Selling Shareholders];
- subject to its obligations under Clause 6 and without prejudice to the underwriting obligations of the Underwriters pursuant to the Underwriting Agreement and this Agreement, [insert name of Joint Bookrunner which is acting as settlement bank] will, subject to Clause 3.1(d) below, be solely responsible for effecting payment for, and delivery of, the Offer Shares in connection with the Offer on behalf of the Underwriters; and
- (d) if agreed with the Joint Bookrunners with respect to such Underwriter, such Underwriter will effect payment to [insert name of Joint Bookrunner which is acting as settlement bank] for and, following receipt of the relevant number of Offer Shares from [insert name of Joint Bookrunner which is acting as settlement bank], will effect delivery of, such number of Offer Shares as may be agreed with the Joint Bookrunners (such number of Offer Shares being Underwriter Settlement Shares) and that such number of Underwriter Settlement Shares may be less than or (with the relevant Underwriter's agreement) greater

The extent to which powers of attorney will be executed by syndicate members (and, therefore, the documents that will be executed by the Joint Bookrunners pursuant to those powers of attorney) to be determined on a case-by-case basis.

In most circumstances, this information will be limited to the Underwriter's name and address although the scope of the information covered may vary from transaction to transaction.

than the number of Shares in its Underwriting Commitment. Any agreement with an Underwriter pursuant to this Clause 3.1(d) will be reached at, or as soon as convenient after, the time at which the Offer Price is set.

- 3.2 The number of Shares to be over-allocated by the Stabilising Manager will be determined by the Joint Bookrunners, on behalf of themselves and each of the other Underwriters. The Stabilising Manager agrees that at no time will the cumulative aggregate number of Shares which it has, on or before the Stabilisation Period End Date, acquired or agreed to acquire in Stabilisation Transactions less the cumulative aggregate number of Shares which it has disposed of or agreed to dispose of in Stabilisation Transactions exceed [15]<sup>10</sup> per cent. of the aggregate Underwriting Commitments of all of the Underwriters.
- 3.3 Subject to the next following sentence, no Underwriter (other than [insert name of Joint Bookrunner which is acting as settlement bank]) will assert any right against the Company [or any Selling Shareholder] or the Joint Bookrunners to require the issue [or transfer] of any Shares to it. If an Underwriter has agreed with the Joint Bookrunners to effect payment for, and delivery of, Offer Shares pursuant to Clause 3.1(d), it will not assert any right against the Company [or any Selling Shareholder] or the Joint Bookrunners to require the issue [or transfer] of more than the number of its Underwriter Settlement Shares.
- 3.4 None of the matters provided for in this Clause 3 will relieve any Underwriter of any of its obligations to the Company [or any of the Selling Shareholders] under clause [] of the Underwriting Agreement, but insofar as:
- there are any Unplaced [New/Sale] Shares, subject to Clause 3.4(b), including as a result of the Joint Bookrunners allocating Shares to the same subscriber or purchaser more than once (or to two or more subscribers or purchasers who have the same investment manager and who intended to make only one order) or, pursuant to Clause 3.1(d), as a result of two or more Underwriters seeking to deliver Underwriter Settlement Shares to the same subscriber or purchaser (or to two or more subscribers or purchasers who have the same investment manager and who intended to make only one order), each Underwriter will be obliged under this Agreement to pay or procure payment for its Agreed Proportion of such Unplaced [New/Sale] Shares; and
- (b) there are Unplaced [New/Sale] Shares which are Underwriter Settlement Shares, any Underwriter which agreed with the Joint Bookrunners to effect payment for, and delivery of, any of such Underwriter Settlement Shares pursuant to Clause 3.1(d) will be obliged under this Agreement to pay or procure payment for such number of those Underwriter Settlement Shares which are Unplaced [New/Sale] Shares if and to the extent that they are Unplaced [New/Sale] Shares by reason of that Underwriter's neglect or default, in each case, in accordance with Clause 6.

## 4. UNDERWRITING

The Joint Bookrunners will promptly inform each Underwriter of the aggregate number of Unplaced [New/Sale] Shares, if any, and of the number of Unplaced [New/Sale] Shares that that Underwriter is required to subscribe for [and purchase] pursuant to the terms of the Underwriting Agreement and this Agreement. Each Unplaced [New/Sale] Share will be subscribed for [or purchased, as the case may be,] by the relevant Underwriter at the Offer Price on the Closing Date (or as soon after the Closing Date as they are so notified by the Joint Bookrunners, as the case may be) and on the terms and subject to the conditions set out, or referred to, in the Underwriting Agreement.

Page 5

This percentage will be determined on a case-by-case basis but it is anticipated that it will be the same percentage that the number of Shares the subject of the over-allotment option bears to the total number of Shares in the Offer. If and to the extent that it is not and a "naked short" is to be permitted, this will need to be agreed by the Stabilising Manager with the Joint Bookrunners. The ability to have a "naked short" may also be affected by the jurisdiction in which the transaction is taking place.

#### 5. **DEFAULTS**

- 5.1 If any Underwriter defaults in its obligation to pay or procure payment for the number of Shares required by Clause 3.4 and Clause 6, the Non-defaulting Underwriter(s) will pay or procure payment for the Defaulted Shares in such proportions as the Joint Bookrunners (or, if one of the Joint Bookrunners is a Defaulting Underwriter, the Joint Bookrunner which is the Non-defaulting Underwriter) may specify **PROVIDED THAT** no Non-defaulting Underwriter will be required as a consequence of the operation of this Clause 5.1, except with its agreement or pursuant to Clause 5.2, to pay or procure payment for more Offer Shares, in aggregate, than the number of Firm Shares in its Underwriting Commitment.
- 5.2 If and to the extent that, following the operation of Clause 5.1 but prior to the operation of this Clause 5.2, there would remain, by virtue of the proviso to that Clause, Defaulted Shares for which no Non-defaulting Underwriter would be required to, or has agreed to, pay or procure payment (*Remaining Defaulted Shares*), the Non-defaulting Underwriters will be severally liable to pay or procure payment for such Remaining Defaulted Shares, by 8.00 a.m. on the day following the Closing Date, in such proportions as the Joint Bookrunners may specify **PROVIDED THAT** no Non-defaulting Underwriter will be required as a consequence of the operation of this Clause 5.2, except with its agreement, to pay or procure payment for a number of Offer Shares which is greater than the sum of:
- (a) the number of Firm Shares in its Underwriting Commitment; and
- (b) the number of Offer Shares that is equal to its Step-up Proportion of the Remaining Defaulted Shares **PROVIDED THAT** in no event shall this number be greater than []<sup>11</sup> per cent. of the aggregate number of Firm Shares for such Non-defaulting Underwriter.
- 5.3 For the avoidance of doubt, the obligations contained in Clause 5.1 and Clause 5.2 are in addition to each Underwriter's payment obligations under Clause 6.1.
- Any Underwriter which fails to pay or procure payment as required by Clauses 3.4, 5.1 or 5.2 hereby authorises the Joint Bookrunners to borrow for its account such sums as may be necessary to permit payment to the Company [and/or the Selling Shareholders (or any of them)] to be effected as required by the Underwriting Agreement and this Agreement, and to pay interest thereon at then current market rates or, as an alternative to borrowing, to lend as principal such sums to, or for the account of, the Defaulting Underwriter, and charge interest thereon at then current market rates. Any amounts (including interest) due from the Defaulting Underwriter as a result of any borrowing or lending pursuant to this Clause 5.4 will be payable to the relevant Non-defaulting Underwriter(s) on demand. Any amounts borrowed pursuant to this Clause 5.4 may be applied by the relevant Non-defaulting Underwriter(s) on behalf of the Defaulting Underwriter for the purposes for which it was borrowed.
- 5.5 If an Underwriter fails to pay or procure payment as required by Clauses 3.4, 5.1 or 5.2, it shall indemnify each Non-defaulting Underwriter against any [additional stamp duty or SDRT or any other similar] tax, charge or duty and any related costs, fines, penalties or interest (*transfer duties*) which becomes payable as a consequence of its default and which is not recoverable under the Underwriting Agreement and against all other losses which each Non-defaulting Underwriter may incur as a direct or indirect result of such failure.
- 5.6 Any Underwriter which defaults in its obligations to pay or procure payment for Offer Shares in accordance with this Agreement will cease to be entitled to any commissions in respect of those Offer Shares to which the failure relates.
- 5.7 The implementation of Clauses 5.1 to 5.6 will not affect the liability of a Defaulting Underwriter in respect of its default, nor relieve it of any of its obligations under this Agreement or the Underwriting Agreement.

Page 6

This percentage will be the maximum percentage step-up on default that is agreed in the default provisions in the underwriting agreement (usually 10%).

5.8 Without prejudice to Clause 5.5 or Clause 5.6, if the Underwriting Agreement is terminated in accordance with clause [] or [] of that Agreement, no Underwriter shall have any additional obligations under Clause 5.1 or Clause 5.2.

#### 6. PAYMENT AND USE OF PROCEEDS

- 6.1 Each Underwriter will:
- (a) pay, or procure payment, to [insert name of Joint Bookrunner which is acting as settlement bank] for value by such time on or after the Closing Date as requested by the [insert name of Joint Bookrunner which is acting as settlement bank] in [pounds sterling] an amount equal to the product of:
  - (i) the Offer Price; and
  - (ii) its Agreed Proportion of any Unplaced [New] Shares [that are not Underwriter Settlement Shares which it or another Underwriter is obliged to pay or procure payment for pursuant to Clause 3.4(b)]; plus
- (b) where it has agreed with the Joint Bookrunners to effect payment for, and delivery of, Underwriter Settlement Shares pursuant to Clause 3.1(d), pay, or procure payment, to [insert name of Joint Bookrunner which is acting as settlement bank] for value by such time on or after the Closing Date as requested by the [insert name of Joint Bookrunner which is acting as settlement bank] in [pounds sterling] an amount equal to the product of:
  - (i) the Offer Price; and
  - (ii) the number of Underwriter Settlement Shares that it is obliged to pay or procure payment for pursuant to Clause [3.4(b)].
- 6.2 [It is acknowledged by the Underwriters and [insert name of Joint Bookrunners which are not acting as settlement bank] that [insert name of Joint Bookrunner which is acting as settlement bank] will receive amounts under Clause 6.1 as agent for and on behalf of the Company [and, where relevant, the Selling Shareholders] and not for its own account.]
- 6.3 [*Insert name of Joint Bookrunner which is acting as settlement bank*] will apply the amounts received by it under Clause 6.1 together with an amount equal to:
- (a) the product of:
  - (i) the Offer Price; and
  - (ii) the number of Offer Shares; less
- (b) the aggregate amount received by it under Clause 6.1, in accordance with the provisions of the Underwriting Agreement and otherwise, without prejudice to any of [insert name of Joint Bookrunner which is acting as settlement bank]'s obligations pursuant to this Agreement, in such manner as it may determine (including, without limitation, in or towards funding Stabilisation Transactions, if any, entered into in accordance with Clause 8).

# 7. SALES OF SHARES

7.1 No Underwriter other than the Stabilising Manager (and any lawful agent appointed by it) will effect any transactions (whether in the open market or otherwise) with a view to stabilising or maintaining the market price of any Security at levels other than those that might otherwise prevail.

- 7.2 Each Underwriter undertakes to each of the Joint Bookrunners that, prior to notification by or on behalf of the Joint Bookrunners that the distribution of the Shares being sold in the Offer has been completed or, if later, the Stabilisation End Date, it will not sell any Offer Shares at less than the Offer Price without the prior consent of the Joint Bookrunners PROVIDED THAT:
- (a) [if an Underwriter is a member of the London Stock Exchange, nothing in this Clause 7.2 shall prevent that Underwriter from trading in Shares on the London Stock Exchange's [Automated Quotation System/Electronic Trading System] in the ordinary course of its business or from facilitating trades (including any associated hedging) on behalf of its clients or from trading in Shares that is otherwise unconnected with the Offer 1<sup>12</sup>; and
- (b) nothing in this Clause 7.2 will restrict the ability of the Stabilising Manager to effect Stabilisation Transactions in accordance with Clause 8.
- 7.3 Except as otherwise provided in this Agreement, no Underwriter will have any duties or obligations of the kind owed to each of their respective clients to any other Underwriter, any affiliate of such Underwriter or any person who acquires Shares in connection with the Offer, by virtue of any communication or the placing of Shares or otherwise.
- 7.4 Except as otherwise provided in this Agreement and in respect of fraud or wilful misconduct, none of the Joint Bookrunners owe any duties or responsibilities nor have any liability to any Non-JBR Underwriter, any affiliate of any Non-JBR Underwriter or any person acquiring Shares in connection with or concerning the allocation of Shares, the establishment of the Offer Price or the suitability of the Shares for a Non-JBR Underwriter, any affiliate of a Non-JBR Underwriter or any person acquiring Shares or otherwise.
- 7.5 Except as expressly provided in this Agreement or in the Underwriting Agreement, but without prejudice to any power of attorney or other authority granted by a Non-JBR Underwriter, none of the Joint Bookrunners will be the agent of any Non-JBR Underwriter, any affiliate of any Non-JBR Underwriter or of any person acquiring Shares in connection with the Offer and no Non-JBR Underwriter will be the agent of any of the Joint Bookrunners or of any associate of any of the Joint Bookrunners or any other Underwriter or of any placee of Shares. The denial of agency by, respectively, the Joint Bookrunners and the Non-JBR Underwriters in this Clause 7.5 includes a denial of authority by any Underwriter to any other Underwriter to make statements or representations on its behalf.

# 8. STABILISATION AND OVER-ALLOTMENT

8.1 On or before the Stabilisation Period End Date, and if and to the extent permitted by applicable laws and regulations, the Stabilising Manager, or its agents, will be entitled (but will not be obliged) to engage in stabilising activities including, without limitation, any activity contemplated by [Article 5 of the Market Abuse Regulation EU 2014/596 (*EU MAR*) and the delegated regulation under EU MAR (EU Regulation 2016/1052) (*the Stabilisation RTS*)]<sup>13</sup> / [Article 5 of the Market Abuse Regulation EU 2014/596 as it forms part of UK domestic law by virtue of the European Union (Withdrawal) Act 2018 (*UK MAR*), the UK Binding Technical Standards for the conditions applicable to stabilisation measures and/or the price stabilising rules made under section 137Q of the FSMA]<sup>14</sup> (any such transactions being referred to in this Agreement as *Stabilisation Transactions*). Stabilisation Transactions, if commenced, may be discontinued at any time

Amend or delete this clause if the Shares are not to be traded in London.

This sentence is the EEA option for use when the shares will be stabilised on an EEA Stock Exchange and/or by an EEA based underwriter. If the securities are traded on an EEA venue (whether regulated market or MTF) EU MAR and the EU stabilisation provisions may apply. Note that UK MAR may potentially apply to in-scope instruments if they are traded on EU venues as well as on UK venues (although for securities on EU venues reporting would be required to relevant EU competent authorities under both UK and EU MAR)

This sentence is the UK option for use when the shares will be stabilised on the London Stock Exchange and/or by a UK based underwriter. It would be included if there is to be stabilisation conducted following the safe harbour set out in Article 5 of the onshored UK Market Abuse Regulation and UK Stabilisation Binding Technical Standards. If the securities being stabilised are

- 8.2 In carrying out Stabilisation Transactions, the Stabilising Manager will act as principal and neither the Stabilising Manager nor any of its agents shall act as agent of any of the other Underwriters. Subject to this Agreement, the exercise of the powers of the Stabilising Manager pursuant to Clause 8.1 (including, without limitation, the decision whether or not to exercise such powers) will be at the absolute discretion of the Stabilising Manager and its agents and neither the Stabilising Manager nor any of its respective employees or agents shall, except in respect of fraud or wilful misconduct or gross negligence, be responsible or liable to, or have any duties to, any of the other Underwriters or any other person in respect thereof including, without limitation, with respect to the timing or terms of any Stabilisation Transaction or the amount of any Stabilisation Profit or Stabilisation Loss (including as a result of any currency exchange or any interest earned on any amount for the account of the other Underwriters). [Notwithstanding the above, the Stabilisation Manager acknowledges and agrees that it will follow any stabilisation strategy that has been or, prior to the Stabilisation Period End Date, is agreed with the Joint Bookrunners.]
- 8.3 The Stabilising Manager will be entitled to use Stabilisation Assets, where necessary after liquidating them, in order to fund Stabilisation Transactions and to meet any costs, expenses or losses it may incur in connection therewith and/or in paying on the Over-allotment Closing Date for any Over-allotment Shares in accordance with paragraph [] of schedule [] to the Underwriting Agreement.
- 8.4 The Stabilising Manager may advance its own funds or securities for the purpose of Stabilisation Transactions. If the Stabilising Manager advances cash, [acting reasonably,] it may charge interest at then current interest rates and if it lends securities it may charge reasonable fees at market rates, and such interest and fees will be taken into account in establishing the amount of the Stabilisation Profit or Stabilisation Loss.
- 8.5 The Underwriters agree that, if and to the extent that the Stabilising Manager acquires Shares under stock lending arrangements in order to ensure that a sufficient number of Shares is available to deliver to investors in connection with the Offer (being a number of Shares equal to the number of Offer Shares), such Shares will be:
- (a) acquired by the Stabilising Manager as principal in accordance with this Clause 8; and
- (b) transferred by the Stabilising Manager to or for the benefit of purchasers procured by the Stabilising Manager in order to satisfy its obligations to deliver Shares to investors in connection with the Offer.

# 9. COMMISSIONS

The Underwriters will be entitled, subject to Clause 5.6 and to Clause 10, to the Offer Commission (together with the applicable VAT thereon), being the aggregate amount payable pursuant to clause [] of the Underwriting Agreement, in the Underwriters' Agreed Proportions<sup>15</sup> [and, by way of discretionary fee, to the aggregate amount payable (if any) pursuant to clause [] of the Underwriting Agreement, in such proportions as the Company [or the Selling Shareholders], as the case may be, may stipulate]<sup>16</sup>.

9.1 "The Underwriters will be entitled, subject to Clauses [5.6, 9.2 and 9.3] and to Clause [10], to the Offer Commission (together with the applicable VAT thereon) on the following basis:

[ by way of *Underwriting Commission*, an amount equal to [ ] per cent, of the aggregate commission payable pursuant to clauses [ ] and [ ] of the Underwriting Agreement (equivalent to [ ] pence per Share), in the Underwriters' Agreed Proportions;

traded only on the UK regulated market (London's main market)), UK MAR and the UK Stabilisation Binding Technical Standards will apply. The Managers will confirm whether the investor base for the issue is such that the securities being stabilised will only be on the London Stock Exchange.

Underwriting, management and selling commissions to be split out as necessary.

If in any particular case it is desirable to split out the underwriting, management and selling commissions consider using the following as a new Clause 9:

- (b) by way of *Management Commission*, an amount equal to [] per cent, of the aggregate commissions payable pursuant to clauses [] and [] of the Underwriting Agreement less [] per cent., which shall be paid as a praecipium to the Joint Global Underwriters' Co-ordinators alone]) (equivalent to [] pence per Share), in the Underwriters' Agreed Proportions;
- (c) by way of *Selling Commission*, an amount equal to [ per cent, of the aggregate commissions payable pursuant to clauses [ ] and [ ] of the Underwriting Agreement (equivalent to pence per Share), in the Underwriters' Agreed Proportions;]
- (d) way of discretionary fee, the aggregate amounts payable (if any) pursuant to clauses [] and [of the Underwriting Agreement, in such proportions as the Company or the Selling Shareholders, as the case may be, may stipulate.]
- 9.2 In relation to Clause [9.1(c)], the Joint Global Co-ordinators agree that their aggregate entitlement to Selling Commissions under that Clause will not exceed [] per cent. of the aggregate of all of the Underwriters entitlements to Selling Commissions under that Clause. If the Joint Global Co-ordinators' entitlement to Selling Commissions is reduced in accordance with this Clause [9.2] the entitlement of the other Underwriters will increase rateably in proportion to their entitlements as established under Clause [9.1] by the amount of the reduction.]
- 9.3 Each of the Underwriters agrees that it will not, without the prior approval of the Joint Global Coordinators, pay or re allow to any person any part of any Selling Commission referred to in Clause [9.1(c)] to which it is entitled."

#### 10. SETTLEMENT OF THE SYNDICATE ACCOUNT

- 10.1 As soon as practicable after the date on which the Stabilising Manager determines that no further Stabilisation Transactions will be entered into and in any event not later than [90] days after the Closing Date, the Syndicate Account will be settled and, subject to Clause 5.6, applied in accordance with the following provisions of this Clause 10. All settlements of the Syndicate Account will be made in [pounds sterling].
- 10.2 Any Stabilisation Profit or Loss will be applied by the Stabilising Manager for the account of each Underwriter subject, in relation to both any Stabilisation Profit and any Stabilisation Loss, to a maximum that is an amount equal to [one times] the amount of Offer Commission that the Underwriter is entitled to pursuant to clause [] of the Underwriting Agreement and Clause 9 of this Agreement (the *Stabilisation Limit*) and, if and to the extent that the amount of any Stabilisation Profit or Stabilisation Loss is less than the Stabilisation Limit, such Stabilisation Profit or Stabilisation Loss will be applied by the Stabilising Manager for the account of each Underwriter in its Agreed Proportion. For the avoidance of doubt, any Stabilisation Profit or Stabilisation Loss in excess of the Stabilisation Limit shall be for the account of the Stabilisation Manager.
- 10.3 [The Joint Bookrunners] may charge each Underwriter's account with any taxes on transfers of Shares made for such Underwriter's account and with such Underwriter's Agreed Proportion of all expenses, liabilities and other obligations howsoever incurred by the Joint Bookrunners on behalf of, or for the accounts of, or for the benefit of, the Underwriters, and with its agreed percentage of any Stabilisation Loss incurred by the Stabilisation Manager, as determined in accordance with Clause 10.2 (the *Agreed Percentage*). The respective accounts of the Underwriters will be settled as set out in this Clause 10, provided that any expenses, liabilities or other obligations not reimbursed or allowed by the Company [or any of the Selling Shareholders] by the time of settlement of the Syndicate Account will be deducted from any Offer Commission payable to the Underwriters. [The Joint Bookrunners] may also reserve and retain such amounts as they may reasonably and in good faith deem necessary or advisable for additional reasonable expenses, liabilities or other obligations not so reimbursed or allowed for by the Company. The Non-JBR Underwriters will not be reimbursed for any of their expenses incurred in connection with the Offer.

- Notwithstanding any settlement by [insert name of Joint Bookrunner which is acting as settlement bank], each Underwriter will remain liable for any taxes on transfers of Shares made for such Underwriter's account and its Agreed Proportion of all expenses, liabilities and other obligations which may be incurred by the Joint Bookrunners on behalf of or for the accounts of themselves and the other Underwriters, as well as its Agreed Percentage.
- 10.5 In the event of default by any Underwriter in respect of its obligations under this Clause 10, each Non-defaulting Underwriter will assume its Step-up Proportion of the obligations of such Defaulting Underwriter pursuant to Clause 5.2 without relieving such Defaulting Underwriter of its liability under this Clause 10.
- 10.6 Insofar as any provision made by the Stabilising Manager in settling the Syndicate Account subsequently proves unnecessary, the amounts reserved will be released to the Underwriters, in the proportions to which they would have been entitled if the provision had not initially been made.
- 10.7 The determination of the Joint Bookrunners, acting in good faith, of any amounts to be paid to or by the Underwriters pursuant to this Clause 10 shall be conclusive and binding on the Underwriters. Any of the Underwriters' funds in the hands of [the Joint Bookrunners] may be held with its general funds without accountability for interest.
- 10.8 [The Joint Bookrunners] shall be entitled to deduct from any monies payable to an Underwriter in accordance with this Clause, or otherwise in connection with the Offer, any amounts payable by that Underwriter pursuant to the Underwriting Agreement or this Agreement, if and to the extent not otherwise already paid.
- 10.9 If and to the extent that the amounts paid by the Company [and/or the Selling Shareholders] to the Joint Bookrunners in respect of any costs, charges, fees and expenses [(or in respect of any associated VAT)] are insufficient to reimburse the Joint Bookrunners in full in respect of the costs, charges, fees and expenses [(or in respect of any associated VAT)] paid or incurred by the Joint Bookrunners in connection with the Offer, each of the Non-JBR Underwriters will pay to, or procure that the same is paid to, the Joint Bookrunners an amount equal to its Agreed Proportion of such un-reimbursed costs, charges, fees and expenses[, together with any associated irrecoverable VAT,] promptly upon receipt of any commissions payable pursuant to this Agreement or, following the settlement of the Syndicate Account, promptly on demand by the Joint Bookrunners.
- 10.10 [Each of the Joint Bookrunners] undertakes to each of the other Underwriters, subject to compliance by each of the other Underwriters with its obligations under Clauses 3 to 6, to account to [HM Revenue & Customs and/or to Euroclear UK & Ireland Limited] for any amounts deducted in respect of transfer duties by [the Joint Bookrunners] from the payments made under the Underwriting Agreement to the Company [and/or the Selling Shareholders] and/or paid by the Company [and/or the Selling Shareholders] to [the Joint Bookrunners] under the Underwriting Agreement in respect of transfer duties (except if and to the extent that any such transfer duties shall have been discharged prior to such deduction or payment or shall not be properly due).

#### 11. REPRESENTATIONS, WARRANTIES AND UNDERTAKINGS

- 11.1 Each of the Underwriters severally, and not jointly and severally, represents and agrees with each of the other Underwriters that:
- (a) it will, in all material respects, comply with such applicable laws in each jurisdiction in which it acquires, offers, sells or delivers any Shares as are customarily complied with by investment banks of international reputation; and
- (b) it will comply with the selling restrictions set out in the Prospectus and Schedule [] of the Underwriting Agreement.
- 11.2 Each Underwriter agrees that, in acquiring, offering, selling or delivering any Shares, it will not rely upon any statement or representation, whether written or oral, express or implied, by any other Underwriter. Each

Underwriter understands and agrees that it is not authorised to give any information or to make any representation not contained in the Prospectus in connection with the acquisition, offer, sale or delivery of Shares.

- 11.3 The undertakings and agreements of each party set out in this Clause 11 shall terminate, upon the earlier of:
- (a) the unanimous agreement of the Underwriters;
- (b) notice being given by or on behalf of the Joint Bookrunners to such effect; and
- (c) 40 days after the Closing Date, unless the Joint Bookrunners have given prior notice that the distribution of the Shares has not yet been completed, in which case the undertakings and agreements shall continue in force until notice to the Non-JBR Underwriters is given by the Joint Bookrunners that the distribution of the Shares has been so completed.

#### 12. TERMINATION

- 12.1 Each party undertakes to each of the other parties to notify the Joint Bookrunners as soon as reasonably practicable if it becomes aware that any of the representations, warranties or undertakings set out in the Underwriting Agreement was, is or has become, or is likely to become, untrue, inaccurate or misleading in any respect or of any other matter or event of the sort referred to in clause [] of the Underwriting Agreement and which, subject to the opinion of such other parties in respect thereof, may give rise to the right to terminate the Underwriting Agreement.
- 12.2 The Joint Bookrunners may, in their absolute discretion, exercise their rights to terminate the Underwriting Agreement.
- 12.3 None of the Joint Bookrunners shall be responsible or liable to any other Underwriter, any affiliate of any party to this Agreement, or any place of Shares or any other person whatsoever for any consequences resulting from the exercise of any discretion referred to in Clause 12.2 (including, without limitation, any termination of, or decision not to terminate, the Underwriting Agreement) or the timing thereof.
- 12.4 Without prejudice to Clause 11.3, upon termination of the Underwriting Agreement for any reason, and in any event on the fortieth day following the Closing Date, the provisions of this Agreement shall terminate, except for this Clause 12 and Clauses 5.5, 7.4, 7.5 and Clauses 14 to 24 without prejudice to any claim relating to any breach of the Underwriting Agreement or this Agreement prior to termination.

#### 13. EXPENSES

All amounts paid by the Company [or the Selling Shareholders] to the Joint Bookrunners under the Underwriting Agreement (or deducted by the Joint Bookrunners from the payments made by the Joint Bookrunners to the Company [and the Selling Shareholders], in accordance with clause [] of the Underwriting Agreement) in respect of any costs, charges, fees and expenses payable by the Company [and/or the Selling Shareholders] under clauses [] and [] of the Underwriting Agreement [or in respect of VAT under clause [] of the Underwriting Agreement] which the Joint Bookrunners may have paid or incurred, shall be retained by the Joint Bookrunners for their own account. If and to the extent that the Joint Bookrunners receive monies from the Company [or the Selling Shareholders] in respect of the costs of one or more of the Underwriters, the Joint Bookrunners shall account to the relevant Underwriter(s) for the relevant proportion thereof. Subject thereto, and save as otherwise provided herein, each of the Underwriters shall bear its own costs and expenses in connection with the Offer, including the negotiation of this Agreement.

#### 14. INDEMNITY

- 14.1 Each party to this Agreement (an *Indemnifying Person*) severally and not jointly or jointly and severally agrees to indemnify and hold harmless each other party and any person who controls such party within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act and such party's and any such person's respective affiliates and subsidiary undertakings and each such person's ultimate parent undertaking and any subsidiary undertaking of such parent undertaking and each of such person's respective directors, officers and employees (each of the foregoing, an *Indemnified Person*) from and against all or any Claims asserted or established against that Indemnified Person in any jurisdiction and against all Expenses which that Indemnified Person may suffer or incur, in each case as incurred, (including, without limitation, all Expenses suffered or incurred in investigating, preparing for or disputing or defending any Claim and/or establishing its right to be indemnified pursuant to this Clause and/or seeking advice regarding any Claim or in any way related to or in connection with this indemnity), if the Claim or Expense arises, directly or indirectly, out of or is attributable to or connected with:
- (a) any of the Offer Documents (or any amendments or supplements thereto) or any other selling or advertising material used by the Underwriters in connection with the Offer, not containing all information required to be contained therein, but, in each case, only with respect to written information furnished by such Indemnifying Person expressly for use in the Offer Documents (or any amendments or supplements thereto) or such other selling or advertising material (such information to be limited to the information set out in Schedule [] to the Underwriting Agreement);
- (b) any untrue statement of a material fact contained in any of the Offer Documents (or any amendments or supplements thereto) or any other selling or advertising material used by the Underwriters in connection with the Offer, or the omission therefrom of a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading, but only with respect to untrue statements or omissions made in the Offer Documents (or any amendments or supplements thereto), or any other selling or advertising material used by the Underwriters in connection with the Offer, in reliance upon and in conformity with written information furnished by such Indemnifying Person expressly for use in the Offer Documents (or any amendments or supplements thereto) or such other selling or advertising material (such information to be limited to the information set out in Schedule [] to the Underwriting Agreement); or
- (c) any breach by such Indemnifying Person of any of the provisions of this Agreement or of the Underwriting Agreement.
- 14.2 Each Indemnified Person will have the right under the Contracts (Rights of Third Parties) Act 1999, to enforce its rights against any Indemnifying Person under this Clause 14 as amended from time to time **PROVIDED THAT** the relevant Underwriter will have sole conduct of any action on behalf of each Indemnified Person connected to it and no enforcement shall be permitted by any Indemnified Person without such Underwriter's consent. The parties to this Agreement may terminate or vary the Agreement in any way and at any time without the consent of any Indemnified Person not a party to this Agreement.
- 14.3 The indemnities contained in this Clause 14 shall remain in full force and effect regardless of any investigations made by or on behalf of any party to this Agreement.
- 14.4 The obligations or liabilities of any Underwriter pursuant to this Clause 14 shall be in addition to any other obligations or liabilities which such Underwriter may have.

# 15. CLAIMS AGAINST UNDERWRITERS

15.1 If any person other than an Underwriter asserts a Claim against one or more of the Underwriters arising out of an untrue statement or omission or alleged untrue statement or omission in the Offer Documents or any of them, each of the Underwriters authorises the Joint Bookrunners, at their election and after such consultation with

the other Underwriters as is reasonably practicable in the circumstances, to make such investigation, to retain such counsel for it and to take such action in the defence of such Claim as the Joint Bookrunners may deem necessary or advisable. Whenever any Underwriter receives notice of the assertion of any Claim to which the provisions of this Clause 15.1 would be applicable, such Underwriter will give prompt notice thereof to the Joint Bookrunners. Whenever the Joint Bookrunners receive notice of the assertion of any such Claim, they will give prompt notice thereof to each Underwriter. The Joint Bookrunners will also furnish each Underwriter with periodic reports, at such times as they deem appropriate, as to the status of any such Claim and the action taken by the Joint Bookrunners in connection therewith.

- 15.2 The Joint Bookrunners will not settle or compromise any Claim:
- (a) in the case of a Claim that is being asserted against only one or some of the Underwriters, without the prior written consent of the relevant Underwriter(s); or
- (b) in the case of a Claim that is being asserted against all of the Underwriters, without the prior written consent of Underwriters holding a majority of the aggregate Underwriting Commitments of all of the Underwriters; or
- (c) unless such settlement or compromise:
  - (i) includes an unconditional release of the Non-JBR Underwriter(s) and/or any other relevant Indemnified Person(s) from all liability arising out of such Claim; and
  - (ii) does not include a statement as to or an admission of fault, culpability or a failure to act by or on behalf of any relevant Indemnified Person,

in which case, no Non-JBR Underwriter or an Indemnified Person of such Non-JBR Underwriter shall have any right, action or claim against the Joint Bookrunners in respect thereof.

- 15.3 Each of the Underwriters will pay its Agreed Proportion of all expenses incurred by the Joint Bookrunners (including the fees and expenses of counsel so retained) in investigating and defending against such Claim and its Agreed Proportion of the aggregate liability incurred by all Underwriters in respect of such Claim (after deducting any contribution or indemnification obtained pursuant to the Underwriting Agreement, or otherwise from persons other than the Underwriters), whether such liability is the result of a judgement against one or more of the Underwriters or the result of any such settlement. A claim against, or liability incurred by, a person who controls an Underwriter shall be deemed to have been made against or incurred by such Underwriter. In the event of default by an Underwriter in respect of its obligations under this Clause 15.3, each Non-defaulting Underwriter shall assume its share of the obligations of such Defaulting Underwriter in its Step-up Proportion without relieving such Defaulting Underwriter of any liability under this Clause 15 or Clause 14.
- 15.4 Notwithstanding Clause 15.1, any Non-JBR Underwriter may retain separate counsel at its own expense.

#### 16. NO PARTNERSHIP

16.1 None of the provisions of this Agreement, the Underwriting Agreement or any other agreement relating to the Offer or the Shares shall constitute or be deemed to constitute a partnership or joint venture between the parties or any of them, or between them (or any of them) and anyone else and, except as specifically provided herein or in the Underwriting Agreement, or as otherwise expressly agreed between the parties in writing, none of the parties shall have any authority to bind any other party in any way. Except as specifically provided herein or in the Underwriting Agreement, no Underwriter shall have any right to contribution or account against any other Underwriter and each Underwriter shall bear all losses and expenses incurred by it and shall be entitled to retain all profits earned by it in connection with the Offer and the issue and sale of Shares.

16.2 If for United States federal income tax purposes this Agreement shall be deemed to create a partnership among two or more of the parties hereto, then each Underwriter hereby elects to be completely excluded from the application of Sub-chapter K, Chapter 1, Subtitle A, of the United States Internal Revenue Code of 1986, as amended, beginning with the first date any such partnership is deemed to have been formed. Each of the Underwriters hereby authorises the Joint Bookrunners, in their absolute unanimous discretion on its behalf, to execute and file with the United States Internal Revenue Service such forms as may be necessary or appropriate to evidence such election.

#### 17. WITHHOLDING AND GROSSING-UP

- 17.1 Save as otherwise provided herein, all sums payable to any of the Underwriters or any other Indemnified Person (for the purposes of this Clause 17 only, each a *payee*) under this Agreement shall be paid free and clear of all deductions or withholdings unless the deduction or withholding is required by law, in which event (and except in the case of interest) the person making the payment (for the purposes of this Clause 17 only the *payer*) shall pay such additional amount as shall be required to ensure that the net amount received by the payee will equal the full amount which would have been received by it had no such deduction or withholding been made.
- 17.2 If the payer makes such an increased payment and the payee subsequently obtains a refund of tax or credit against tax by reason of the payer making such a deduction or withholding, the payee shall reimburse the payer as soon as reasonably practicable with such amount as the payee shall determine to be such proportion of the said refund or credit as shall leave the payee after such reimbursement in no better or worse position (having regard to the time value of money) than it would have been in had no deduction or withholding been required. Nothing in this Clause 17.2 shall oblige the payee to disclose to the payer, nor shall the payer be entitled to inspect, any of the books and other records of the payee nor shall anything herein prevent the payee from arranging its tax and commercial affairs in whatever manner it thinks fit and, in particular, the payee shall not be under any obligation to claim credit or relief from or against its corporate profits or similar liability to tax in respect of the amount of such deduction or withholding as aforesaid in priority to any other reliefs available to it.
- 17.3 If [HM Revenue & Customs or] any [other] tax authority brings into charge to tax (or into any computation of income, profit or gains for the purposes of any charge to tax) any sum payable to a payee under this Agreement or any sum withheld in accordance with this Agreement from any payment made to a payee (other than in respect of payments made under Clause 10, the Offer Commission referred to in Clause 9 and payments of interest[, in each case which are subject to corporation tax on the payee's actual net income, profits or gains]) the person liable under this Agreement to make such payment shall pay such additional amount as shall be required to ensure that the total amount received by the payee, less the tax chargeable thereon (or that would be so chargeable but for the availability of relief in respect of that charge to tax), is equal to the amount that would otherwise be so received (additional payments being made on demand of the payee).
- 17.4 In calculating the liability of any person to any payee pursuant to the indemnities contained in Clause 14, there shall be taken into account, having regard to the time value of money, the amount by which any liability to tax of the payee (or any person affiliated with the payee) is actually reduced or extinguished as a result of the matter giving rise to the indemnity claim.

#### 18. INFORMATION PROCEDURES

- 18.1 Each of the Non-JBR Underwriters undertakes to each of the Joint Bookrunners to provide to the Joint Bookrunners, in such form and at such times as it may reasonably request, any other information which it may reasonably request in connection with the Offer.
- 18.2 Each of the Non-JBR Underwriters undertakes to each of the Joint Bookrunners to notify it as soon as practicable if it becomes aware that it has breached this Agreement or that the provisions of Clause 4 have become relevant.

#### 19. ADVERTISING

- 19.1 Each of the Underwriters is authorised to send such ordinary course Bloomberg messages or "external use" e-mails or such other ordinary course communications, targeted at permitted professional investors, as are necessary or appropriate in relation to marketing the Offer on behalf of the Company [and the Selling Shareholders] on such dates and in such countries as it shall determine, in its absolute discretion, **PROVIDED THAT** any such Bloomberg message, "external use" e-mail or other ordinary course communication complies with applicable laws and regulations and the terms of this Agreement and the Underwriting Agreement. Each Underwriter consents to any necessary or appropriate use of its name in any such Bloomberg message, "external use" e-mail or other ordinary course communication provided that the name appears in the form and context previously notified to each of the other Underwriters.
- 19.2 Save to the extent contemplated in Clause 19.1, each of the Non-JBR Underwriters represents and warrants to each of the other Underwriters that it has not released any advertisement, announcement or other similar communication to the public (including, without limitation, press releases, general advertising or any "tombstone" advertisements) of, or in relation to, the Offer, its role in the Offer, the Group or the Shares in any jurisdiction (other than, for the avoidance of doubt, case studies that are targeted at such Underwriter's investment banking clients only), and undertakes to each of the other Underwriters that it will not do so prior to the Stabilisation Period End Date or, if earlier, the date on which all the New Shares [and Sale Shares] have been taken up or placed or, failing which, all the Unplaced New Shares [and Unplaced Sale Shares] have been subscribed for [or purchased] by the Underwriters pursuant to the terms of the Underwriting Agreement and this Agreement, without the prior consent of the Joint Bookrunners.
- 19.3 Without prejudice to Clause 19.2, each party undertakes that any such advertisement, announcement or other similar communication to the public (as referred to in Clause 19.2) that it may make, and any case studies that it may issue, will comply with applicable laws and regulations and will be made in all respects on its own behalf and at its own expense and that no other party shall have any liability therefor or arising therefrom.

# 20. NO AMENDMENTS OR MODIFICATIONS EXCEPT IN WRITING

This Agreement may not be amended or modified except in writing executed by each of the parties hereto<sup>17</sup> save that each Underwriter (whether it is an original party to this Agreement or a party to whom this Agreement has been previously transferred pursuant to this paragraph) shall be entitled to assign or transfer all of its rights or obligations under this Agreement to any affiliate registered in the European Union or which is also carrying on EU-regulated services (in each case, the "EU Affiliate") by notice in writing, and from the date of such transfer, references to such Underwriter shall be read as references to such EU Affiliate. Upon completion of such assignment or transfer of rights and obligations under this Agreement, each transferor pursuant to this Clause 20 shall be released from its obligations but not its rights under this letter agreement].

21. [ONLY INSERT THIS CLAUSE IF, IN THIS AGREEMENT, EEA UNDERWRITERS ARE CONTRACTING UNDER NON-EEA LAW OR UK UNDERWRITERS ARE CONTRACTING UNDER NON-ENGLISH LAW] - ACKNOWLEDGMENT AND CONSENT TO BAIL-IN OF FINANCIAL INSTITUTIONS<sup>18</sup>

Delete the wording following "hereto" to the end of the sentence, unless exceptionally an Underwriter expects to have to transfer its role to an EU affiliate before the transaction will close and settle.

The UK left the EU on 31 January 2020 and the transition period during which the UK was no longer part of the EU and not a Member State, but still subject to EU law, ended on 31 December 2020. This means that, for EEA Underwriters, English law is now a third country law, and so EEA Underwriters who enter English law-governed agreements among underwriters and underwriting agreements are required to include the BRRD Article 55 contractual recognition language, requiring their counterparties' acknowledgement and consent to the bail-in provisions required under EU law. A good example of such wording is set out in Clause 21.1 of this AFME Form AAU.

UK Underwriters who enter an agreement among underwriters or an underwriting agreement governed by a non-English law are required by the UK Prudential Regulation Authority to include UK-specific recovery and resolution clauses in such agreements, requiring their counterparties' acknowledgement and consent to the bail-in provisions required under English law. A good example of such wording is set out in Clause 21.2 of this AFME Form AAU.

- 21.1 [Alternative A Form of wording for use by EEA Underwriters in contracts governed by a non-EEA law] Notwithstanding and to the exclusion of any other term of this Agreement or any other agreements, arrangements or understanding between the Underwriters, each of the Underwriters acknowledges and accepts that a BRRD Liability arising under this Agreement may be subject to the exercise of Bail-in Powers by the Relevant Resolution Authority, and acknowledges, accepts, and agrees to be bound by:
- (a) the effect of the exercise of Bail-in Powers by the Relevant Resolution Authority in relation to any BRRD Liability of an EEA Underwriter to each other Underwriter under this Agreement, that (without limitation) may include and result in any of the following, or some combination thereof:
  - (i) the reduction of all, or a portion, of the BRRD Liability or outstanding amounts due thereon;
  - (ii) the conversion of all, or a portion, of the BRRD Liability into shares, other securities or other obligations of the relevant EEA Underwriter or another person (and the issue to or conferral on the other Underwriters of such shares, securities or obligations);
  - (iii) the cancellation of the BRRD Liability;
  - (iv) the amendment or alteration of any interest, if applicable, thereon, the maturity or the dates on which any payments are due, including by suspending payment for a temporary period;
- (b) the variation of the terms of this Agreement, as deemed necessary by the Relevant Resolution Authority, to give effect to the exercise of Bail-in Powers by the Relevant Resolution Authority.
- 21.2 [Alternative B Form of wording for use by UK Underwriters in contracts governed by non-English law]<sup>19</sup> Notwithstanding and to the exclusion of any other term of this Agreement or any other agreements, arrangements, or understanding between the Underwriters, each of the Underwriters acknowledges and accepts that a UK Bail-in Liability arising under this Agreement may be subject to the exercise of UK Bail-in Powers by the relevant UK resolution authority, and acknowledges, accepts, and agrees to be bound by:
- (a) the effect of the exercise of UK Bail-in Powers by the relevant UK resolution authority in relation to any UK Bail-in Liability of a UK Underwriter to each other Underwriter under this Agreement, that (without limitation) may include and result in any of the following, or some combination thereof:
  - (i) the reduction of all, or a portion, of the UK Bail-in Liability or outstanding amounts due thereon;
  - (ii) the conversion of all, or a portion, of the UK Bail-in Liability into shares, other securities or other obligations of the relevant UK Underwriter or another person, and the issue to or conferral on the other Underwriters of such shares, securities or obligations;
  - (iii) the cancellation of the UK Bail-in Liability;
  - (iv) the amendment or alteration of any interest, if applicable, thereon, the maturity or the dates on which any payments are due, including by suspending payment for a temporary period;

The source of this example bail-in wording for EEA and UK Underwriters (which the example bail-in wording in this Agreement mirrors) is the "Recognition of UK Bail-in Clause for Other Liabilities" section of AFME and ICMA's 13 November 2020 document entitled "Recognition of EU and UK Bail-in Clauses for Other Liabilities".

If either an EEA Underwriter or a UK Underwriter enters an agreement among underwriters or an underwriting agreement governed by a non-EEA and non-English law (e.g. Swiss or NY law), both forms of bail-in and contractual recognition language referenced above will need to be included in such agreement. The relevant wording in clause 21.1/2 of this AFME Form AAU may be used for this purpose.

(b) the variation of the terms of this Agreement, as deemed necessary by the relevant UK resolution authority, to give effect to the exercise of UK Bail-in Powers by the relevant UK resolution authority.

#### 22. MISCELLANEOUS

- 22.1 20 [Alternative A Form of wording for EEA Banks who are subject directly to the EU MiFID II Product Governance regime] In relation to the Offer and solely for the purposes of Article 9(8) of Commission Delegated Directive 2017/593 (the "Delegated Directive") regarding the responsibilities of Manufacturers under the Product Governance requirements contained within: (a) Directive 2014/65/EU on markets in financial instruments, as amended ("MiFID II"); (b) Articles 9 and 10 of the Delegated Directive; and (c) local implementing measures (the "MiFID II Product Governance Requirements"), each [Joint Global Coordinator/Joint Bookrunner note: the exact group of "manufacturers" is to be determined on case by case deal facts and circumstances] acknowledges to each other [Joint Global Coordinator/Joint Bookrunner] that it understands the responsibilities conferred upon it under the MiFID II Product Governance Requirements relating to: (i) the target market for the Offer; (ii) the eligible distribution channels for dissemination of the Offer Shares, each as set out in the [Prospectus and the] [Announcement] [to be] dated [•] in relation to the Offer; and (iii) the requirement to carry out a product approval process.
- 22.2 [Alternative B Form of wording for UK Banks who are subject to the UK onshored Product Governance regime] In relation to the Offer and solely for the purposes of Paragraph 3.2.7R regarding the responsibilities of UK Manufacturers under the Product Governance requirements contained within Chapter 3 of the FCA Handbook Product Intervention and Product Governance Sourcebook (the "UK Product Governance Requirements"), each [Joint Global Coordinator/Joint Bookrunner note: the exact group of "manufacturers" is to be determined on case by case deal facts and circumstances] acknowledges [to each other [Joint Global Coordinator/Joint Bookrunner]] that it understands the responsibilities conferred upon it under the UK Product Governance Requirements relating to: (i) the target market for the Offer; (ii) the eligible distribution channels for dissemination of the [securities the subject of the Offer], each as set out in the [Prospectus] [Announcement] [to be] dated [•] in relation to the Offer; and (iii) the requirement to carry out a product approval process.
- 22.3 (a) In the event that any Underwriter that is a Covered Entity becomes subject to a proceeding under a U.S. Special Resolution Regime, the transfer from such Underwriter of this Agreement, and any interest and obligation in or under this Agreement, will be effective to the same extent as the transfer would be effective under the U.S. Special Resolution Regime if this Agreement, and any such interest and obligation, were governed by the laws of the United States or a state of the United States.
- (b) In the event that any Underwriter that is a Covered Entity or a Covered Affiliate of such Underwriter becomes subject to a proceeding under a U.S. Special Resolution Regime, Default Rights under this Agreement that may be exercised against such Underwriter are permitted to be exercised to no greater extent than such Default Rights could be exercised under the U.S. Special Resolution Regime if this Agreement were governed by the laws of the United States or a state of the United States.

This note applies equally to Clause 22.1 and 22.2. Depending on the location of the manufacturers, there may be situations where either the MiFID II product governance language or the UK MiFIR product governance language is included or, more rarely, where both are included. Clauses 22.1 and/or 22.2 should include all entities deemed to be EU MiFID or UK MiFIR Manufacturers in the relevant offering. This should be considered on a case by case basis with respect to all MiFID or UK MiFIR entities deemed to be manufacturers in relation to the shares, and will vary depending on the facts of the relevant offering/which MiFID or UK MiFIR entities are collaborating with the issuer in the creation, development, issue and/or design of the shares which (as described in the ESMA Technical Advice of 19 December 2014) includes entities "advising corporate issuers on the launch of the new securities" (i.e. liaising with the issuer in structuring the launch, book-build and allocation of the new securities). In some cases (for example where the Joint Global Co-ordinators are the entities substantively collaborating with the Issuer), it may be appropriate for the Joint Global Co-ordinators to be considered as co-manufacturers; in other cases the group of co-manufacturers may extend to some or all of the Joint Bookrunners. Consider whether the issuer and/or any Financial Adviser will also be a co-Manufacturer (particularly in cases where the issuer is a regulated entity).

- 22.4 The obligations of each of the parties under this Agreement are separate, not joint or joint and several and, accordingly, each party shall (save as otherwise agreed among them) have the right to protect and enforce its rights without joining the other parties in any proceedings.
- 22.5 Except as expressly otherwise provided in this Agreement [and/or the Underwriting Agreement], the taking of any action, making of any decision or exercise of any discretion pursuant to this Agreement by, and any determination made by, the Joint Bookrunners shall be on a joint and unanimous basis.
- 22.6 Each party to this Agreement agrees that it has not relied on any, and has acted independently of each other party to this Agreement in determining to execute and be a party to this Agreement and/or the Underwriting Agreement.
- 22.7 [Save as set out in Clause 14.2, a person who is not a party to this Agreement shall have no right under the Contracts (Rights of Third Parties) Act 1999 to enforce any terms of this Agreement, but this does not affect any right or remedy of a third party which exists or is available apart from that Act.]
- 22.8 Time shall be of the essence of this Agreement, both as regards the times, dates and periods mentioned herein and as regards any times, dates and periods which may, by agreement in writing between the parties hereto, be substituted for them.
- 22.9 Interest shall run (before and after judgement) on any sums due and payable hereunder from the due date for payment thereof until the date of actual payment at the Agreed Rate.
- 22.10 If and to the extent that the operation of any provision of this Agreement results in a number of Shares which includes a fraction of a Share, such fraction shall be rounded either up or down to the nearest whole number, or dealt with in such manner as the Joint Bookrunners shall determine.
- 22.11 [Notwithstanding the terms of the Underwriting Agreement, in relation to any discretion in the Underwriting Agreement that requires the unanimous consent of [the Joint Bookrunners], in the event that there is not unanimous consent, the agreement of [a majority in number of [the Joint Bookrunners / Non-defaulting Underwriters holding a majority of the aggregate Underwriting Commitments of all of the Non-defaulting Underwriters]] shall be deemed to be unanimous consent and, accordingly, the unanimous consent of [the Joint Bookrunners] shall then be notified in accordance with the procedures set out in the Underwriting Agreement for the exercise of the relevant discretion.]<sup>21</sup>
- 22.12 For the avoidance of doubt, the provisions of this Agreement apply in place of the ICMA limits on stabilisation losses and unreimbursed expenses as set out in the ICMA Primary Market Handbook, and such limits are hereby expressly disapplied. In addition, unless otherwise agreed in writing by the Underwriters, none of the other ICMA recommendations as set out in the ICMA Primary Market Handbook shall apply to the Offer

#### 23. PARTIAL INVALIDITY OR UNENFORCEABILITY

The invalidity or unenforceability for any reason whatsoever of any provision of this Agreement or the Underwriting Agreement will in no way affect the validity or enforceability of any of the other provisions of this Agreement or the Underwriting Agreement or of any other agreements or arrangements of which this Agreement or the Underwriting Agreement forms part, which provisions will remain valid and enforceable in all respects.

Page 19

This wording is only likely to be used in limited circumstances and where agreed to, on a case-by- case basis, by the Joint Bookrunners. Care should be taken to consider any risks that might arise (and the advisability of including the clause or a similar clause) if it was deemed with hindsight to be inconsistent with the terms of the Underwriting Agreement as agreed with the Company [and the Selling Shareholders]. Alternative decision-making procedures may be agreed by the Joint Bookrunners.

#### 24. EXECUTION AND ADHERENCE

- 24.1 This Agreement may be executed by any one or more of the parties hereto in any number of counterparts, each of which shall be deemed to be an original, but all such counterparts shall together constitute the same instrument. Delivery of a counterpart of this Agreement by e-mail attachment or fax shall be an effective mode of delivery.
- 24.2 Execution of this Agreement by or on behalf of any party to this Agreement will constitute a warranty and undertaking to each other party to this Agreement that such party's affiliates have complied, and that each party to this Agreement will procure that its affiliates will comply, in all respects with the provisions of this Agreement as if they were a party to this Agreement.

#### 25. NOTICES

- 25.1 Save as otherwise provided in this Agreement, any notice, demand or other communication to be served under this Agreement shall be in writing and shall be served upon any party hereto only by posting by first class post (if to an address in the same country) or air mail (if to an address in a different country) or delivering the same to its or his address given or referred to in this Clause 23 or sending the same by facsimile transmission to the number given in this Clause 23 for the addressee or at such other address or number as it or he may from time to time notify in writing in accordance with this Clause 23 to the other parties hereto or by email.
- A notice or demand served by first class post shall be deemed duly served on an addressee in the same country forty-eight hours (disregarding days which are not Business Days) after posting, a notice or demand served by air mail shall be deemed duly served on an addressee in a different country seven days (disregarding days which are not Business Days) after posting and a notice or demand delivered personally or sent by facsimile transmission shall be deemed duly served at the time of delivery or transmission (save that if the delivery or transmission occurs after 6.00 p.m. on a Business Day or on a day which is not a Business Day the notice or demand shall be deemed duly served at 8.30 a.m. on the next Business Day following delivery or transmission) and in proving service of the same it will be sufficient to prove, in the case of a letter, that such letter was left at or delivered to the correct address of the party to be served as provided in this Agreement or in the case of properly stamped or franked first class post or air mail, addressed to the address of the party to be served given in this Clause 23 and placed in the post and, in the case of a facsimile transmission, that such facsimile was duly transmitted to the number of the party to be served given in this Clause 23 and an electronic acknowledgement was received. A notice or demand sent by email shall be deemed served upon the generation of a receipt notice by the recipient's server or, if such notice is not so generated, upon delivery to the recipient's server.
- 25.3 All notices, demands or other communications given under this Agreement, shall be given to the following addresses and numbers:

If to [insert name of Global Co-ordinator One]:	[ ] [ ] [ ]		
	For the attention of: [ ] Fax		
	Number: [	] Tele	phone
	Number: [	]	
	Email: [ ]		

If to [insert name of Global Co-ordinator Two]:	[ ] [ ] [ ]
	For the attention of: [ ] Fax
	Number: [ ] Telephone
	Number: [ ]
	Email: [ ]
If to [insert name of Underwriter One]:	[ ] [ ] [ ]
	For the attention of: [ ] Fax
	number: [ ] Telephone
	number: [
	Email: [ ]
If to [insert name of Underwriter Two]:	[ ] [ ] [ ]
	For the attention of: [ ]
	Fax number: [ ]
	Telephone number: [ ]
	Email: [ ]
If to [insert name of Underwriter Three]:	[ ] [ ] [ ]
	For the attention of: [ ] Fax
	number: [ ] Telephone
	number: [
	Email: [ ]

#### 26. GOVERNING LAW AND JURISDICTION

- 26.1 This Agreement and the relationship among the parties to it and any non-contractual obligations which may arise out of or in connection with this Agreement shall be governed by and interpreted in accordance with English law.
- All the parties agree that the courts of England are to have non-exclusive jurisdiction<sup>22</sup> to settle any dispute (including claims for set-off and counter claims) which may arise out of or in connection with (i) the creation, validity, effect, interpretation, or performance of, or of legal relationships established by, this Agreement or otherwise arising out of or in connection with this Agreement, and (ii) any non-contractual obligations which may arise out of or in connection with this Agreement, and for such purposes irrevocably submit to the non-exclusive jurisdiction of the English courts<sup>23</sup>.
- 26.3 The parties irrevocably agree that a judgement and/or order of any court referred to in this Clause based on any matter arising out of or in connection with this Agreement is to be conclusive and binding on it and may be enforced against it in the courts of any other jurisdiction, whether or not (subject to due process having been served on it) it participates in the relevant proceedings.
- [Each of the parties with an address outside England shall at all times maintain an agent for service of process and any other documents and proceedings in England or any other proceedings in connection with this Agreement. For each of the Non-JBR Underwriters with an address outside England, such agent shall be the person identified in Schedule 2. Any writ, judgment or other notice of legal process shall be sufficiently served on the relevant party if delivered to such agent at its address for the time being. Each of the parties with an address outside England irrevocably undertakes not to revoke the authority of the above agent and if, for any reason, the Joint Bookrunners (for themselves or on behalf of the Indemnified Persons) request such party to do so it shall promptly appoint another such agent with an address in England and advise each of them. If, following such request, the relevant party fails to appoint another agent, the Joint Bookrunners shall be entitled to appoint one on the relevant party's behalf and at such party's expense.]

**IN WITNESS WHEREOF** this Agreement has been duly executed under hand by the parties hereto.

Under a non-exclusive jurisdiction clause, syndicate member banks are able to bring a direct claim against another party in the local court, join another party in proceedings in the local court and may be able bring a contribution claim against another party in a local court (if the concept exists locally), as well as bring a direct claim in the English courts and establish a due diligence defence (if the concept exists locally).

As an alternative, standard LCIA arbitration clause with an option to litigate may be agreed by the parties, especially in agreements involving banks located in jurisdictions where arbitration may be more appropriate.

#### **SCHEDULE 1**

#### **DEFINITIONS AND INTERPRETATION**

1. In this Agreement the following words and expressions shall have the meanings set out below:

Agreed Percentage has the meaning given in Clause 10.3;

**Agreed Proportion** means, in relation to Global Co-ordinator One, [] per cent., in relation to Global Co-ordinator Two, [] per cent., in relation to Underwriter One, [] per cent., in relation to Underwriter Two, [] per cent., and in relation to Underwriter Three, [] per cent.;

Agreed Rate means [the Bank of England's Bank Rate as published by the Bank of England from time to time];

[affiliate means [if there is no definition of "affiliate" in the Underwriting Agreement, appropriate definition to be included]];

*Bail-in Legislation* means in relation to any member state of the European Economic Area which has implemented, or which at any time implements, the BRRD, the relevant implementing law, regulation, rule or requirement as described in the EU Bail-in Legislation Schedule from time to time.

[Bail-in Powers means any Write-down and Conversion Powers as defined in the EU Bail-in Legislation Schedule, in relation to the relevant Bail-in Legislation.]

[BRRD means Directive 2014/59/EU establishing a framework for the recovery and resolution of credit institutions and investment firms.]

[BRRD Liability means a liability in respect of which the relevant Write-down and Conversion Powers in the applicable Bail-in Legislation may be exercised]

**Claim** means any and all claims, actions, liabilities, demands, proceedings, investigations, judgments or awards whatsoever and including any commenced or threatened investigation or proceeding by any governmental agency or body (in each case whether or not successful, comprised or settled and whether joint or several);

Closing Date means [] or such other date (not later than [] (the Long Stop Date)) for settlement of subscriptions and sales, as the case may be, under the Offer as the Company and the Joint Bookrunners (for themselves and on behalf of the other Underwriters) may agree; [To track definition in the Underwriting Agreement]

Co-Lead Managers means Underwriter Two and Underwriter Three;

*Company* has the meaning given in Recital (A);

Covered Affiliate has the meaning assigned to the term "affiliate" in, and shall be interpreted in accordance with, 12 U.S.C. § 1841(k);

Covered Entity means any of the following:

- (i) a "covered entity" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 252.82(b);
- (ii) a "covered bank" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 47.3(b); or

(iii) a "covered FSI" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 382.2(b);

**Default Right** has the meaning assigned to that term in, and shall be interpreted in accordance with, 12 C.F.R. §§ 252.81, 47.2 or 382.1, as applicable:

**Defaulted Shares** means Offer Shares in respect of which an Underwriter has failed to pay or procure payment for as required by Clause 3.4 and Clause 6;

**Defaulting Underwriter** means an Underwriter which fails to pay or procure payment in accordance with this Agreement;

[*EEA Underwriter* means an Underwriter incorporated and regulated in a member state of the European Economic Area.]

[EU Bail-in Legislation Schedule means the document described as such, then in effect, and published by the Loan Market Association (or any successor person) from time to time at http://www.lma.eu.com/pages.aspx?p=499.]

**Expense** means any and all losses, damages, costs, demands, charges or expenses (including legal fees) and Taxes (other than corporation tax incurred by the relevant Indemnified Person on its actual net income, profits or gains);

EU Prospectus Regulation means the Prospectus Regulation EU 2017/1129;

*Firm Shares* means the aggregate number of Shares in respect of which the Underwriters are obliged, pursuant to clause [] of the Underwriting Agreement, to procure subscribers [or purchasers] for, or failing which, themselves to subscribe for [or purchase];

**Indemnified Person** has the meaning given in Clause 14.1;

*Indemnifying Person* has the meaning given in Clause 14.1;

ICMA means the International Capital Market Association;

ICMA Primary Market Handbook means the Handbook published by ICMA;

Joint Bookrunners means the Joint Global Co-ordinators [and [ ]];

Joint Global Co-ordinators means Global Co-ordinator One and Global Co-ordinator Two;

Long Stop Date has the meaning given to it in the definition of Closing Date in this Schedule 1;

[Management Commission has the meaning given in Clause 9.1(b);]<sup>24</sup>

Non-defaulting Underwriter means an Underwriter which is not a Defaulting Underwriter;

Non-JBR Underwriters means Underwriter Two and Underwriter Three;

Offer has the meaning given in Recital (A);

See footnote 13. Only include this definition if in any particular case it is desirable to split out the underwriting, management and selling commissions and the alternative Clauses set out in footnote 13 are included in the Agreement.

*Offer Commission* means the aggregate commissions payable pursuant to clause [] of the Underwriting Agreement;

Offering Circulars has the meaning given in Recital (A);

*Offer Shares* means the aggregate number of Firm Shares plus any Shares over-allotted in connection with the Offer;

party means a party to this Agreement and parties shall be construed accordingly;

Pathfinder Prospectus has the meaning given in Recital (A);

**Prospectus** has the meaning given in Recital (A);

[Relevant Resolution Authority means the resolution authority with the ability to exercise any Bail-in Powers in relation to any EEA Underwriter.]

**Remaining Defaulted Shares** has the meaning given in Clause 5.2;

Securities means any Shares, any securities or other instruments convertible into or exchangeable for or representing rights to acquire or dispose of Shares and any securities convertible into or exchangeable for Shares, and includes any contract for differences or other derivative relating to any securities or any arrangement which otherwise comprises Securities and Security shall be construed accordingly;

[Selling Commission has the meaning given in Clause 9.1(c);]25

**Shares** means ordinary shares of [] pence each in the capital of the Company;

*Stabilisation Assets* shall comprise such amounts as are determined by the Stabilising Manager but shall include (without limitation):

- (a) the sums received or receivable by the Stabilising Manager under Clause 6.1 insofar as they exceed the sum of the amounts payable by it under Clause 6.2 (otherwise than in respect of Over-allotment Shares);
- (b) any Securities held or acquired as a result of any Stabilisation Transaction;
- (c) all income received or earned in respect of such Securities; and
- (d) any cash proceeds arising out of Stabilisation Transactions;

*Stabilisation Liabilities* shall comprise such amounts as are determined by the Stabilising Manager but shall include (without limitation):

- (a) all sums payable to lenders (including, if applicable, the Stabilising Manager or any other Underwriter) in respect of interest, fees or otherwise, on any funds or securities advanced to or borrowed by the Stabilising Manager for the purposes of Stabilisation Transactions (including in order to settle over-allotments in the Offer);
- (b) all other payments, costs and expenses (including tax and transfer duty liabilities, costs and expenses) made or incurred in connection with Stabilisation Transactions (including in respect

See footnote 13. Only include this definition if in any particular case it is desirable to split out the underwriting, management and selling commissions and the alternative Clauses set out in footnote 13 are included in the Agreement.

of the cost of acquiring any Securities or rights arising as a result of Stabilisation Transactions) and any loans or other temporary acquisitions of any Securities (together, *Stabilisation Expenses*); and

(c) all provisions made by the Stabilising Manager, when establishing the Syndicate Account, for any of the amounts referred to in sub-paragraph (a) or (b) above,

and [excluding], for the avoidance of doubt, any amounts payable to the Over-allotment Shareholders in respect of Over-allotment Shares pursuant to paragraph [] of schedule [] to the Underwriting Agreement.

Stabilisation Limit has the meaning given in Clause 11.2;

Stabilisation Loss means, where the Stabilisation Assets are less than the Stabilisation Liabilities, the amount of such deficit;

Stabilisation Profit means, where the Stabilisation Assets exceed the Stabilisation Liabilities, the amount of such excess;

Stabilisation Transaction has the meaning given in Clause 8.1;

*Step-up Proportion* means, in respect of a Non-defaulting Underwriter, the proportion which its Underwriting Commitment bears to the aggregate Underwriting Commitments of all Non-defaulting Underwriters;

Syndicate Account means such account or accounts designated as such by [the Joint Bookrunners];

**[UK Bail-in Legislation** means Part I of the UK Banking Act 2009 and any other law or regulation applicable in the UK relating to the resolution of unsound or failing banks, investment firms or other financial institutions or their affiliates (otherwise than through liquidation, administration or other insolvency proceedings).]

[UK Bail-in Liability means a liability in respect of which the UK Bail-in Powers may be exercised.]

*[UK Bail-in Powers* means the powers under the UK Bail-In Legislation to cancel, transfer or dilute shares issued by a person that is a bank or investment firm or affiliate of a bank or investment firm, to cancel, reduce, modify or change the form of a liability of such a person or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability.]

*UK Prospectus Regulation* means the Prospectus Regulation EU 2017/1129 as it forms part of UK domestic law by virtue of the European Union (Withdrawal) Act 2018;

[UK Underwriter means an Underwriter incorporated and regulated in the United Kingdom.]

*U.S. Special Resolution Regime* means each of (i) the U.S. Federal Deposit Insurance Act and the regulations promulgated thereunder and (ii) Title II of the U.S. Dodd-Frank Wall Street Reform and Consumer Protection Act and the regulations promulgated thereunder;

*underwrite* means an obligation of an Underwriter arising under the Underwriting Agreement to subscribe for New Shares [and/or to purchase, Sale Shares] in the event that the Underwriters fail to procure subscribers and purchasers for all of the New Shares [and Sale Shares];

Underwriters means the Joint Global Co-ordinators, Underwriter One and the Co-Lead Managers;

*Underwriter Settlement Shares* has the meaning ascribed to it in Clause 3.1(d);

*Underwriting Agreement* has the meaning given in Recital (B);

[Underwriting Commission has the meaning given in Clause 9.1(a);]<sup>26</sup>

*Underwriting Commitment* means, in respect of a particular Underwriter, the number of Firm Shares calculated by applying the percentage set opposite its name in column [] of the table in Part [] of schedule [] to the Underwriting Agreement to the maximum number of Firm Shares;

Unplaced New Shares has the meaning ascribed to it in Recital (B); [and] [Unplaced Sale Shares has the meaning ascribed to it in Recital (B); [and]] [Unplaced Shares has the meaning ascribed to it in Recital (B).]

- 2. References in this Agreement to any Recital, Clause or Schedule are to recitals of, clauses of, and schedules to, this Agreement (except where the context otherwise requires).
- 3. Headings in this Agreement shall not affect its interpretation.
- 4. In this Agreement, words incorporating the singular only shall include the plural and vice versa and references to *persons* shall include individuals, bodies corporate (wherever incorporated), unincorporated associations and partnerships (including limited partnerships) and wherever any such associations or partnerships are formed or organised.
- 5. References to any statute, regulation or part thereof in this Agreement shall be construed as references thereto as amended or re-enacted (whether before or after the date of this Agreement) or as the application thereof is modified by other provisions prior to the date hereof or subsequently (provided that no modification subsequent to the date of this Agreement shall increase or extend the liability of the Underwriters under this Agreement), shall be construed as including references to any provision of which they are re-enactments (whether with or without modification) and shall be construed as including references to any order, instrument, regulation or other subordinate legislation made pursuant thereto.
- 6. References in this Agreement to times and dates are to London times and dates.
- 7. Each reference in this Agreement to a breach of any of its provisions includes a reference to any of the representations or warranties given in this Agreement being inaccurate or misleading.
- 8. Each reference to the terms and conditions of the Offer are to those terms and conditions as set out in the relevant Offer Document and as implied by law, regulation and any applicable market usage.
- 9. Unless the context otherwise requires, each reference in this Agreement to each of the Joint Bookrunners or any Non-JBR Underwriter by any description or in any capacity includes a reference to it in each other capacity in which it may act pursuant to this Agreement or the Underwriting Agreement.

Page 27

See footnote 13. Only include this definition if in any particular case it is desirable to split out the underwriting, management and selling commissions and the alternative Clauses set out in footnote 13 are included in the Agreement.

# SCHEDULE 2

# [AGENT FOR SERVICE OF PROCESS]

[Insert details where relevant.]

SIGNED by for and on behalf of [GLOBAL CO-ORDINATOR ONE]	)
SIGNED by for and on behalf of [GLOBAL CO-ORDINATOR TWO]	)
SIGNED by as attorney for and on behalf of [UNDERWRITER ONE]	)
SIGNED by as attorney for and on behalf of [UNDERWRITER TWO]	)
SIGNED by as attorney for and on behalf of [UNDERWRITER THREE]	)

[ ] 20[]

[ ]PLC

# AGREEMENT AMONG UNDERWRITERS

# **CONTENTS**

CLAUSE		PAGE
1.	INTERPRETATION	2
2.	AUTHORITIES	3
3.	BOOKBUILDING AND OFFERING	4
4.	UNDERWRITING	5
5.	DEFAULTS	6
6.	PAYMENT AND USE OF PROCEEDS	7
7.	SALES OF SHARES	7
8.	STABILISATION AND OVER-ALLOTMENT	8
9.	COMMISSIONS	9
10.	SETTLEMENT OF THE SYNDICATE ACCOUNT	10
11.	REPRESENTATIONS, WARRANTIES AND UNDERTAKINGS	11
12.	TERMINATION	12
13.	EXPENSES	12
14.	INDEMNITY	13
15.	CLAIMS AGAINST UNDERWRITERS	13
16.	NO PARTNERSHIP	14
17.	WITHHOLDING AND GROSSING-UP	15
18.	INFORMATION PROCEDURES	15
19.	ADVERTISING	16
20.	NO AMENDMENTS OR MODIFICATIONS EXCEPT IN WRITING	16
21.	[ONLY INSERT THIS CLAUSE IF, IN THIS AGREEMENT, EEA UNDERWRITERS ARE	

	UNDER NON-EEA LAW OR UK UNDERWRITERS ARE CONTRACTING UNDER NON-ENGLISH LAW] - ACKNOWLEDGMENT AND CONSENT TO BAIL-IN OF FINANCIAL INSTITUTIONS
22.	MISCELLANEOUS
23.	PARTIAL INVALIDITY OR UNENFORCEABILITY 19
24.	EXECUTION AND ADHERENCE
25.	NOTICES
26.	GOVERNING LAW AND JURISDICTION
SCHEDI	JLE 1 DEFINITIONS AND INTERPRETATION
SCHED	JLE 2 [AGENT FOR SERVICE OF PROCESS]28